Author’s response to reviews


Authors:

Ki Wook Yun (pedwilly@gmail.com)
Hoan Jong Lee (hoanlee@snu.ac.kr)
Jin Han Kang (kjhan@catholic.ac.kr)
Byung Wook Eun (acet0125@hanmail.net)
Yae-Jean Kim (yaejean.kim@gmail.com)
Kyung-Hyo Kim (kaykim@ewha.ac.kr)
Nam Hee Kim (peddream@medimail.co.kr)
Young Jin Hong (hongyjin@inha.ac.kr)
Dong Ho Kim (kdh281920@naver.com)
Hwang Min Kim (khm9120@yonsei.ac.kr)
Sung-Ho Cha (sunghocha@khu.ac.kr)

Version: 5
Date: 24 July 2014

Author’s response to reviews:

July 24, 2014

Editor-in-Chief
BMC Infectious Diseases
Re: MS: 1554678346122581


Dear Professor Philippa Harris,

Thank you very much for your 2nd review of our revised manuscript. Please find our responses to the expert reviewers’ comments. We appreciate each comment and have made a series of revisions that we believe have much improved the manuscript. Please do not hesitate to contact me if you have further questions or concerns.

Sincerely,

Hoan Jong Lee, MD
Professor of Pediatrics
Seoul National University College of Medicine
Seoul National University Children's Hospital
101 Daehak-ro, Jongno-gu Seoul, 110-769, Republic of Korea
Tel: 82-2-2072-3633
Fax: 82-2-745-4703
E-mail: hoanlee@snu.ac.kr